Author's response to reviews

Title: Do smartphone applications in healthcare require a governance and legal framework? It depends on the application!

Authors:

Esmita Charani (e.charani@imperial.ac.uk)
Enrique Castro-Sánchez (e.castro-sanchez@imperial.ac.uk)
Luke SP Moore (l.moore@imperial.ac.uk)
Alison Holmes (alison.holmes@imperial.ac.uk)

Version:3 Date: 14 January 2014

Author's response to reviews:

Dear Editors,

Many thanks for accepting our commentary. We have now according to your recommendations made the following final changes to the document:

1. Removed the sections from the abstract to confirm with the guidelines. The abstract is 168 words.

2. Removed the subheading of ‘discussion’ from the main body of text.

We look forward to hearing from you,

Sincerely

Esmita